Dear Editor of BMC Public Health

Please consider this paper for publication in the Biomedical Journal of Public Health.

We believe that the Biomedical Journal of Public Health is the most appropriate place to publish this work as it is open access so that other researchers and practitioners can freely access this paper.

Yours sincerely

Karla Canuto
PhD Candidate
University of South Australia
Adelaide